C

O

S

4

2 3 , A p

r i l 4,1

99 6

H a

n d o u t Nu

m

b

er 9H

o

m e

wor

k Se

t

No

.

4

A nn

o

un c

e men t

T

he next homework

set

(not this one, but Homework Set

No.

5

) will be ap

roject, in the sense that everyone should do it by oneself

without collaboration. Rea d in g Assignments

Read

[

K ] Chapters 22-24 and [CLR] Chapter 37

; read

[CLR] Chapter 36 and [G

J] Chpater 3 as needed for doing the homeworks.Optiona l Readings

W e plan to give some lectures on Cook’s theorem and

maybe counting problems for\cultural enhancement".

Y ou may want to

read [GJ] Chapter 2 and [K] Chapters 25-27 if you want to understand

thoroughly the material covered in these lectures.Written

Assignments

Part

I

{

In the book [GJ], on pages 75-76, 14 problems are listed:

A

.

Prove that Problem 3 is NP-complete.B.

Prove that Problem 5 is NP-complete.C.

Prove that Problem 7 is NP-complete.D.

Prove that Problem 11 is NP-complete.E.

Prove that Problem 12 is NP-complete.

Part II { In the book [CLR]:F.

Do Problem 36-1 (page 961).G.

DoExercise 37.2-2 (page 973).

G. Do Problem 37-1 (page 983).